Author’s response to reviews

Title: Meat consumption providing a surplus energy in modern diet contributes to obesity prevalence: an ecological analysis

Authors:

Wenpeng You (wenpeng.you@adelaide.edu.au)

Maciej Henneberg (maciej.henneberg@adelaide.edu.au)

Version: 3 Date: 21 Mar 2016

Author’s response to reviews:

The following has been included in the cover letter to address the comments from the editor:

Therefore, please add a 'Declarations' heading to your manuscript after the 'Conclusions' section. This declarations section should contain the following sub-sections:

- List of abbreviations

  If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided.

  Authors: All the abbreviations were used in the text have been defined in the text at their first use. A list of the abbreviations has been included in the “Declarations”.

- Ethics (and consent to participate): please move the statement from the Methods section to appear in the Declarations.

  Authors: We move the statement from the Methods section to the appear in the Declarations section.

- Consent to publish: please state “Not applicable” in this section.

  Authors: “Not applicable” has been stated.

- Competing interests (present; no addition required)

  Authors: The previous “Competing interests” statement is kept as it is.

- Funding

  All sources of funding for the research reported should be declared. The role of the funding body in the design of the study and collection, analysis, and interpretation of data
and in writing the manuscript should be declared. If no funding was obtained for this study, please state so in this section.

Authors: There was no funding for this project. This statement has been included in the Funding section as well.

- Authors' contributions (present; no addition required)

Authors: The previous “Authors’ contribution” statement is kept as it is.

- Availability of data and materials

Please detail where the data supporting your findings can be found (including information found in any supplementary files). If you do not wish to share your data, please state (in the ‘Availability of data and Materials’ section) that data will not be shared, and state the reason. If all the data supporting your findings is contained within the manuscript, please state so in the “Availability of data and materials” section. Please note that identifying/confidential patient data should not be shared.

Authors: The following description has been added in the section of “Availability of data and materials”:

All data for this research were freely downloaded from the United Nations agents’ websites. The sources and data robustness have been described in the section of “Methods”. Furthermore, all the data supporting our findings is contained within the Supplemental File titled, Table S2 Detailed information of country-level estimates.

- Acknowledgements (present; no addition required)

Authors: The previous “Acknowledgements” statement is kept as it is.